Author's response to reviews


Authors:

Fatma N Al-Maskari (fatma.am@uaeu.ac.ae)
Mohammed El-Sadig (msadig@uaeu.ac.ae)

Version: 3 Date: 9 March 2007

Author's response to reviews: see over
In response to reviewer R Tapp

Major compulsory revisions:

1. Agreed and fixed
2. Agreed and fixed
3. Fixed
4. Fixed